

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Sola, Alessandro

Working Paper The 2015 refugee crisis in Germany: Concerns about immigration and populism

SOEPpapers on Multidisciplinary Panel Data Research, No. 966

Provided in Cooperation with: German Institute for Economic Research (DIW Berlin)

Suggested Citation: Sola, Alessandro (2018) : The 2015 refugee crisis in Germany: Concerns about immigration and populism, SOEPpapers on Multidisciplinary Panel Data Research, No. 966, Deutsches Institut für Wirtschaftsforschung (DIW), Berlin

This Version is available at: https://hdl.handle.net/10419/178208

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



SOEPpapers

on Multidisciplinary Panel Data Research

SOEP - The German Socio-Economic Panel study at DIW Berlin

966-2018

5668

The 2015 Refugee Crisis in Germany: Concerns about Immigration and Populism

Alessandro Sola



SOEPpapers on Multidisciplinary Panel Data Research at DIW Berlin

This series presents research findings based either directly on data from the German Socio-Economic Panel study (SOEP) or using SOEP data as part of an internationally comparable data set (e.g. CNEF, ECHP, LIS, LWS, CHER/PACO). SOEP is a truly multidisciplinary household panel study covering a wide range of social and behavioral sciences: economics, sociology, psychology, survey methodology, econometrics and applied statistics, educational science, political science, public health, behavioral genetics, demography, geography, and sport science.

The decision to publish a submission in SOEPpapers is made by a board of editors chosen by the DIW Berlin to represent the wide range of disciplines covered by SOEP. There is no external referee process and papers are either accepted or rejected without revision. Papers appear in this series as works in progress and may also appear elsewhere. They often represent preliminary studies and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be requested from the author directly.

Any opinions expressed in this series are those of the author(s) and not those of DIW Berlin. Research disseminated by DIW Berlin may include views on public policy issues, but the institute itself takes no institutional policy positions.

The SOEPpapers are available at http://www.diw.de/soeppapers

Editors: Jan Goebel (Spatial Economics) Stefan Liebig (Sociology) David Richter (Psychology) Carsten Schröder (Public Economics) Jürgen Schupp (Sociology)

Conchita **D'Ambrosio** (Public Economics, DIW Research Fellow) Denis **Gerstorf** (Psychology, DIW Research Fellow) Elke **Holst** (Gender Studies, DIW Research Director) Martin **Kroh** (Political Science, Survey Methodology) Jörg-Peter **Schräpler** (Survey Methodology, DIW Research Fellow) Thomas **Siedler** (Empirical Economics, DIW Research Fellow) C. Katharina **Spieß** (Education and Family Economics) Gert G. **Wagner** (Social Sciences)

ISSN: 1864-6689 (online)

German Socio-Economic Panel (SOEP) DIW Berlin Mohrenstrasse 58 10117 Berlin, Germany

Contact: soeppapers@diw.de



The 2015 Refugee Crisis in Germany: Concerns about Immigration and Populism^{*}

Alessandro Sola[†]

February 2018

Abstract

This paper investigates the effect of the refugee crisis, and the related government's asylum policy, on concerns about immigration of the German population. Exploiting exogenous variation in survey interview timing of the German Socio-Economic Panel (SOEP), I employ a difference-in-differences strategy to estimate the short-term causal effect of the refugee crisis on concerns about immigration. The estimated effect is substantial, representing an increase in concerns of around 22%, compared to the pre-refugee crisis baseline level. Interestingly, I find that this increase was twice as large for East Germans, compared to West Germans. In a second section, I show concerns about immigration are positively correlated with political support for the relatively new, right-wing populist party *Alternative für Deutschland* (AfD). However, using the variability in concerns on political preferences in the short term.

JEL Classification: F22, J61, D72, H12.

Keywords: concerns about immigration, refugee crisis, Germany, AfD, populism, political preferences.

*I am grateful to my supervisior Guido Tabellini for his advice and guidance. I also thank Timothy Hatton and Christian Dustmann for helpful suggestions. All errors are my own.

[†]Email: alessandro.sola@hotmail.it.

1 Introduction

The refugee crisis sparked heated political debates throughout Germany, exposing polarised attitudes toward immigration. In 2015 alone, around 890,000 people entered the country (BAMF, 2016). This unprecedented arrival of asylum seekers and migrants uncovered divergent views on how to handle the crisis and on its potential impact on German society. For some people, the seemingly unstoppable influx represented a threat and a cause for major concern. At the same time, the *Alternative für Deutschland* ("Alternative for Germany", AfD), a relatively new, right-wing populist party, sustained a rising anti-immigration and anti-refugee rhetoric, tapping into people's anxieties. In the federal election of 2017, the AfD achieved a remarkable electoral success becoming the third largest party in the German parliament.

Did the refugee crisis substantially increase German concerns about immigration? And if so, was this increase homogenous across the whole German population? More importantly, are these concerns associated with the recent success of right-wing populism in Germany? This paper attempts to answer these questions, also providing other interesting findings.

In this paper, I utilise data from the German Socio-Economic Panel (SOEP), a nationally representative longitudinal survey, interviewing annually around 30,000 individuals. Interviews are conducted randomly throughout the year and the dataset provides a wide range of information at the individual level. Exploiting exogenous variation in survey interview timing of the SOEP, I employ a difference-in-differences strategy to estimate the short-term causal effect of the refugee crisis, and the related government's asylum policy, on concerns about immigration. The treatment group is composed of all individuals interviewed during the peak of the refugee crisis, from 1 June until the end of the year, in 2015. The causal effect is identified by comparing the difference in outcomes of the treatment group in 2015 and 2014, with the same difference for the control group. The empirical strategy hinges on the crucial identifying assumption that, in the absence of the refugee crisis, the outcomes of the treatment and control group would have followed parallel trends. I provide evidence of the credibility of this assumption by showing that the estimated coefficients of placebo treatment interactions are close to zero and statistically insignificant in the years prior to the treatment.

I estimate that the refugee crisis, and the related permissive government's asylum policy, increased the share of individuals who are very concerned about immigration to Germany by 6 percentage points. This represents a substantial effect and constitutes an increase of around 22% relative to the pre-refugee crisis, baseline level of concerns.

Consistent with the literature on anti-immigration preferences, I find that the level of education is strongly negatively correlated with concerns about immigration. I show these results are robust to alternative specifications and that the refugee crisis increased concerns about hostility towards foreigners and minorities. Interestingly, I test the heterogeneity of the treatment effect, finding that the refugee crisis increased concerns about immigration of East Germans more than double that those of West Germans.

In the last part of the paper, I investigate the link between concerns about immigration to Germany and support for the AfD, a right-wing populist party with a strong anti-immigration stance. In individual fixed effects regressions, I find that concerns about immigration are positively correlated with support for the AfD, and not with support for all the other parties. Using the variation in concerns generated by the refugee crisis, I employ an instrumental variable strategy to provide tentative evidence of the effect of concerns on support for the AfD. I do not find any evidence of a causal effect of concerns on political preferences. However, these results present some limitations. More specifically, it is probable that individuals do not fully disclose their real preference, especially when it comes to a right-wing populist party expressing radical and xenophobic positions. Moreover, my analysis focuses on the short-term and it is possible that the AfD mobilised large shares of the electorate closer to the election day.

The first part of the paper contributes to the vast literature on public opinion and attitudes toward immigration. Most studies in economics and political science have focused on the determinants and consequences of anti-immigration preferences, analysing individual views on immigration or immigrants through survey measures. Hainmueller and Hopkins (2014) provide a comprehensive review of this literature, marking a distinction between the political economy strand, mostly focused on individual self-interest explanations, and the sociopsychological tradition, which stresses the role of group-related attitudes and symbols.¹ Among the former strand, Scheve and Slaughter (2001) find that less-skilled workers are significantly more likely to hold restrictive preferences over immigration to the United States. They measure labour market skills by the average occupation wages and the years of education completed, and these findings are consistent with the view that individuals hold immigration-related opinions based on their interests as labour force participants. This interpretation is further confirmed by crosscountry data, as high-skilled individuals are less likely to be opposed to immigration in countries where the skill composition of natives relative to immigrants is high (Mayda, 2006; O'Rourke and Sinnott, 2006).

Instead, Hainmueller and Hiscox (2007) provide an alternative interpretation of the

¹For other reviews of the literature, see Ceobanu and Escandell (2010) and Hatton (2016a).

role of education, finding that its effect on attitudes is related to differences in individual values and beliefs. They argue that more educated individuals have less restrictive preferences over immigration because they are more tolerant, open to cultural diversity, and are more likely to believe immigration is good for the host economy. The importance of cultural factors is also estimated with a latent-factor model for 21 European countries by Card *et al.* (2012). They find that concerns about compositional amenities are 2 to 5 times more important than concerns about wages and taxes in explaining variability of anti-immigration preferences. Individuals seem to value being surrounded by people who share the same ethnicity, language, norms and traditions, but the level of education weakens this preference.

However, recent experimental evidence by Haaland and Roth (2017) suggests that labour market concerns play a quantitatively significant role in determining attitudes toward immigration. Exploiting exogenous variation in people's beliefs about the impact of immigration on the labour market, they find a persistent change in attitudes and a related shift in political behaviour. Moreover, another strand of the literature has emphasised the role played by the media on anti-immigration preferences, depending on whether immigration is portrayed in a positive or negative manner (Facchini *et al.*, 2009; Boomgaarden and Vliegenthart, 2009; De Poli *et al.*, 2017).

From a methodological standpoint, the closest paper to mine is Schüller (2016), who finds that the 9/11 attacks substantially increased concerns about immigration, and decreased concerns about hostility toward foreigners in Germany.² Moreover, a recent paper by Hatton (2017) stresses the need to account for the salience of immigration, defined as the importance which is attached by individuals to immigration, relative to other policy issues. I complement this strand of the literature by uncovering the correlations between measures of preference, salience and concerns about immigration of the German population. I provide evidence suggesting that my measure of concerns from the SOEP, although associated to both, is more correlated to a measure of preferences than to a measure of salience, as represented by European Social Survey and Eurobarometer data respectively.

The second part of the paper is related to two strands of the literature. First, a series of studies have analysed the possible causes of the recent success of populism in the Western world, ranging from the economic crisis and globalization shocks, to a decrease of trust in institutions and a cultural backlash (Algan *et al.*, 2017; Rodrik,

²For other papers employing similar difference-in-differences strategies with SOEP data, see Goebel *et al.* (2013), Caliendo and Wrohlich (2010) and Doerrenberg and Siegloch (2014).

2017; Dustmann *et al.*, 2017b; Inglehart and Norris, 2016).³ Moreover, Guiso *et al.* (2017) find that economic insecurity plays a direct effect on populist voting and a causal effect on attitudes toward immigration. They argue that anti-immigration preferences are not an autonomous determinant of populist voting, but rather a channel through which economic insecurity affects political preferences. I argue that, for the case of Germany, people's concerns about immigration may be an independent driver of support for populist parties. In fact, Germany is the country that received the most refugees in Europe and was not affected by the economic crisis as much as others.

Second, a recent body of research has analysed the link between immigration and voting for centre-right or far right parties (Halla *et al.*, 2017; Barone *et al.*, 2016; Dustmann *et al.*, 2016). Among these, Otto and Steinhardt (2014) provide evidence of the causal relationship between immigration and the success of far-right parties at the city district level in Hamburg. In my paper, directly investigating the change in public opinion, I attempt to uncover a possible channel trough which a consistent influx of refugees might affect political preferences. More research, covering more recent years, is needed to investigate whether the increase in concerns about immigration generated by the refugee crisis shifted political support towards the AfD.

The rest of the paper is organised as follows. Section 2 provides background information about the refugee crisis, the government's asylum policy and their effect on concerns about immigration. Section 3 describes the data and relates the dependent variable of concerns to other measures commonly used in the literature. Section 4 explains the empirical strategy employed for identification. Section 5 presents the empirical results, performs robustness checks, and analyses the heterogeneity of the treatment effect. Section 6 links concerns about immigration to support for the AfD. Finally, Section 7 concludes.

2 Background

2.1 The 2015 Refugee Crisis

An unprecedented influx of asylum seekers and migrants made their way to Europe in 2015, marking the largest refugee crisis since the Second World War. Over 1.2 million first-time asylum applications were lodged in the EU member states in 2015, with Germany being the first destination country.⁴ Such vast movements of people were mostly

³For a review on populism, see Gidron and Bonikowski (2013).

 $^{^{4}}$ First-time asylum applications count people who lodged an application for asylum for the first time in an EU member state. However, it is possible that a person applies again in another member state.

a consequence of the series of revolutions and civil wars in North Africa and the Middle East, commonly known as the "Arab Spring".

The crisis reached its peak in Germany in the second half of 2015. In Figure 1, blue bars represent the number of arrivals of people who intended to apply for asylum in 2014–2017, registered through the EASY system, an IT application for the initial registration and distribution of asylum seekers to German states.⁵ Orange bars represent the number of first-time asylum applications received in Germany (Eurostat data). Starting from June 2015, the number of registrations in the EASY system was increasing at a higher pace than the number of asylum applications. The German public administration could not cope with the rising influx of asylum seekers, most of which could lodge an application several months after their arrival in the country (Singleton *et al.*, 2016). In the same figure, the dashed red line reports the salience of immigration as a policy issue for the German population (Eurobarometer data).⁶ This variable closely follows the flow of arrivals of asylum seekers, reaching its peak in late 2015.

At the end of August 2015, according to article 17 of the Dublin III Regulation, Germany decided to examine asylum claims of Syrian citizens, without sending them back to the country of first entry.⁷ On 31 August, German Chancellor Angela Merkel, during her speech at the Federal Press Conference, announced emphatically: "Wir schaffen das" ("We can do it"), effectively committing to a permissive asylum policy. Some days later, she further stressed that there would be no legal limit to the number of refugees Germany would be accepting. Her message resonated around the world, signalling migrants the possibility to successfully seek asylum in Germany. Thus, the temporary asylum policy change (i.e., the suspension of the Dublin procedures), coupled with Merkel's speech, constituted a "pull" effect for asylum seekers and migrants.

In fact, asylum applications are determined by country of origin factors, like genocide, civil war or human rights abuse, and by country of destination variables, like high income, good employment prospects and the restrictiveness of asylum policies. Hatton (2016b) estimates the determinants of asylum applications to 19 OECD countries over 1997–2012, uncovering the deterrent effects of though asylum policies.⁸ It is possible that the

⁵EASY statistics might be inflated due to double-counting and registration errors. While the total number of registered arrivals in 2015 is almost 1.1 million people, the Federal Office for Migration and Refugees (BAMF) estimated a more realistic figure of 890,000. Starting from 2017, the EASY system registers personal information of the asylum seekers, thus avoiding any miscalculation.

 $^{^{6}\}mathrm{Information}$ on the survey question underling this measure is given in Section 3.2.

⁷EU Regulation No 604/2013 (Dublin III Regulation) establishes that the EU member state responsible for examining the asylum claim is the first country in which the asylum seeker lodged its application. (http://eur-lex.europa.eu/LexUriServ/LexUriServ.do?uri=OJ:L:2013:180:0031:0059:EN:PDF). As of 21 October 2015, Germany put the standard procedures back into place.

 $^{^{8}}$ The toughness of asylum policies is measured through an index composed of 15 factors, divided



Figure 1: Arrivals of asylum seekers and salience of immigration in Germany

Source: BMI (2017), Eurostat (2017), Eurobarometer (2017). Notes: Blue bars represent the number of arrivals of people who intent to apply for asylum in Germany, registered through the EASY system. Orange bars represent first-time asylum applications lodged in Germany. The dashed red line reports the salience of immigration as a policy issue for the German people (scale on the right axis). It is calculated as the share of respondents that report "Immigration" as one of the two most important issues facing Germany, in the Eurobarometer survey.

more permissive asylum policy and Merkel's rhetoric further increased arrivals of asylum seekers and migrants in Germany, and in Europe in general.

However, it is difficult to prove rigorously this is the case in such a short time frame, as the decision to migrate depends from exogenous factors in the country of origin we are not properly able to control for. Instead, I can provide descriptive evidence suggesting that, after Merkel's speech, Germany was receiving a higher proportion of asylum seekers among all the other European Union countries. In Figure 2, I plot the monthly share of extra-EU first-time asylum applications lodged in Germany, out of all extra-EU first-time asylum applications in the EU, in 2012–2017. After Merkel's speech, the German share drastically increased from an average level of 29% over the previous years to 66% in February 2016. Considering Syrian asylum seekers only, the share increased even more sharply, reaching 90% in February 2016.⁹ Nevertheless, to the extent that German welcoming policies at the end of 2015 did not increase arrivals in Europe, a higher share of

in the 3 following categories: characteristics related to the access to the hosting country, aspects that determine the procedures for recognition of asylum, and the welfare provisions during and after the processing of an application.

 $^{^{9}}$ In 2015, Syrian nationals made up 36% of all first-time asylum applications to Germany.



Figure 2: Share of extra-EU first-time asylum applications lodged in Germany, out of all EU28 countries

Notes: The figure plots the monthly share of extra-EU first-time asylum applications lodged in Germany, out of all extra-EU first-time asylum applications in the EU, in the period 2012-2017. The first red line represents 31 August 2015, the day of Merkel's speech ("We can do it"). The second red line represents 18 March 2016, the day of the EU-Turkey Statement.

the asylum seekers that would have come anyway to the EU, was attracted to Germany.

On 18 March 2016, the EU signed a statement with Turkey intending to limit the incentives to migrate. Any irregular migrant entering the EU through Turkey could be sent back, while the EU would agree to resettle in Europe any migrants qualified for asylum in the same proportion. The agreement also envisaged financial aid to Turkey for the refugees hosted there and the acceleration of visa liberalization for Turkish nationals. Following the deal, the arrival of migrants from the Balkan route greatly decreased (blue bars in Figure 1). Instead, asylum seekers already residing in the country kept lodging applications, as they could not do so in the previous months (orange bars in Figure 1). Once the number of asylum applications started going down in October 2016, so did the share received by Germany (Figure 2). After the peak of the crisis, and with a lag of a couple of months after the EU-Turkey Statement due to the slow lodging of applications, the German share returned to its baseline level of around 30%.

In sum, the peak of the refugee crisis in Germany was reached in the second half of 2015. The government enacted a permissive asylum policy, while the German Chancellor encouraged a welcoming culture, possibly further increasing arrivals. Starting from June 2015, the number of registrations in the EASY system was increasing at a higher pace

Source: Eurostat (2017).

than the number of asylum applications. After Merkel's speech at the end of August, the German share of extra-EU first-time asylum applications increased sharply.

2.2 Effect of the Crisis on Concerns about Immigration

The refugee crisis sparked heated political debates throughout the country and uncovered polarised attitudes toward migrants and refugees, with divergent views on their potential impact on society. On the one hand, there was a surge in the number of hostile acts against refugees, among which personal injuries, arson attacks on refugee accommodations and anti-immigrant demonstrations. Most of the violent acts took place in East Germany and there is a correlation between these events and the vote shares for extreme right and populist right-wing parties at the distinct level (Jäckle and König, 2017). On the other hand, the "welcome politics" sustained by the Chancellor were embraced by a vast number of people who volunteered enthusiastically to help the arrival and integration of refugees into the German society.

The crisis was characterized by a seemingly unstoppable inflow of asylum seekers and migrants, who were perceived to be culturally more distinct from previous immigrants (Dustmann *et al.*, 2017a). Such a consistent arrival of refugees can constitute a contextual determinant causing the manifestation of preferences toward immigration and an increase in the salience of the policy issue. A variety of theories account for restrictive preferences toward immigration at the individual level, ranging from economic to cultural explanations. First, self-interested individuals could hold negative attitudes based on their concerns about the perceived impact of immigration on wages and taxes. Second, the arrival of refugees could threaten the national identity and more basic prejudicial reasons would form the basis for opposition toward immigration. Moreover, migration flows and the number of asylum applications are positively correlated with the salience of immigration as a policy issue (Hatton, 2017a). Thus, the refugee crisis has likely increased natives' concerns about immigration.

As noted, the second half of 2015 marked the peak of arrivals, possibly further spurred by the welcoming policy enacted by the German government. Hence, I estimate the increase in concerns from 1st June until the end of the year.¹⁰ The treatment I analyse in this paper consists of both the refugee crisis and the permissive government's asylum policy. The management of the crisis in August 2015 changed the expectations of asylum seekers and migrants regarding the way Germany would handle their claims. But most importantly for my analysis, it changed the expectations of the German people.

 $^{^{10}\}mathrm{In}$ a robustness check, I utilize the day after Merkel's speech of 31st August as the start of the treatment.

Indeed, natives' concerns are likely influenced both by current (and perceived) migration flows and by the future expectation about these flows. Even if the permissive asylum policy – epitomised by Merkel's speech – did not affect actual inflows, it might have changed Germans' perceptions about how many more people would have been arriving to the country. To be sure, to the extent that the government's stance did not affect asylum seekers and migrants' expectations, and German concerns, what I estimate is solely the effect of the peak of the refugee crisis.

3 Data

In Section 3.1, I present the dataset and basic descriptive statistics. In Section 3.2, I analyse the association of my dependent variable of concerns with alternative indicators of public opinion used in the literature. I find that my variable is more correlated to a measure of preference, than to one of salience of immigration.

3.1 Dataset and Descriptive Statistics

The empirical analysis of this paper utilizes data from the German Socio-Economic Panel (SOEP), a nationally representative longitudinal survey, interviewing annually around 12,000 households and nearly 30,000 individuals since 1984 (Wagner *et al.*, 2007). The dataset contains a wide range of information at the individual level, including socioeconomic characteristics, labour-market outcomes and a variety of attitudinal measures. Interviews are conducted randomly throughout the year, from January until November.¹¹ Thus, there is substantial variability at the individual level within and across years. For instance, a given individual might be interviewed in February once, and in September the following year. This exogenous variation in survey interview timing allows me to identify the short-term effect of the refugee crisis.

The dependent variable I utilise in the main analysis is derived from answers to the following question: "How concerned are you about the following issues?". Respondents answer on several issues, namely "The economy in general", "Your own economic situation", "Your health", "Immigration to Germany", "Hostility toward foreigners or minorities in Germany", and others. Respondents can answer "very concerned", "somewhat concerned", and "not concerned at all". My dependent variable is a dummy equal to one for individuals who are very concerned about immigration to Germany, and zero otherwise. In 2014 (2015), 25.32% (31.64%) of respondents were very concerned about it,

 $^{^{11}{\}rm Most}$ of the interviews are conducted in the first half of each year. For instance, in my dataset, 77% of the interviews in 2015 are conducted until the end of June.



Figure 3: Daily average of "Very concerned about immigration" in 2015

Notes: The figure plots the daily average of the dependent variable, a dummy equal to one if the respondent is "very concerned about immigration to Germany", and zero otherwise. The red line represents 1 June 2015, the start of the treatment.

41.80% (42.36%) were somewhat concerned, and 32.87% (26.00%) were not concerned at all. In Figure 3, I plot the daily average of the dependent variable for 2015. It is evident that the share of individuals that are very concerned about immigration to Germany increases sharply after June, during the peak of the refugee crisis.

The analysis is conducted dividing the full sample of respondents in 2015 in a treatment and a control group. The treatment group is composed of all individuals who were interviewed from June 2015 until the end of the year.¹² All these individuals were exposed to the treatment, namely the peak of the refugee crisis and the related government's asylum policy. Accordingly, the control group is composed of all individuals interviewed between 1 January and 30 May 2015. It is important to note that this group division is time-invariant (where time is measured in years), because it depends on interview timing in 2015.

The dataset covers all individuals interviewed in 2012–2015. I exclude respondents who were not interviewed in 2015, as I would not be able to divide them between the treatment and control group. I also exclude individuals with missing values for my dependent variable. Below, my main analysis includes only respondents for 2014 and

Source: SOEP 2016.

¹²The last interview contained in the dataset was conducted on 10 November.

	20	014		2	015	
	Control	Tretment	Norm. Diff.	Control	Tretment	Norm. Diff.
Ν	16,398	7,134		16,845	8,308	
Immigration Concerns	0.255	0.249	0.010	0.300	0.349	-0.074
Age	51.54	42.17	0.416	51.93	41.91	0.434
Age 26-35	0.135	0.168	-0.065	0.130	0.148	-0.037
Age 36-45	0.172	0.296	-0.209	0.169	0.269	-0.172
Age 46-64	0.344	0.325	0.028	0.341	0.341	0.000
Age 65 over	0.269	0.067	0.396	0.279	0.066	0.415
Male	0.465	0.442	0.033	0.466	0.443	0.033
Secondary Education	0.614	0.575	0.056	0.615	0.583	0.046
Tertiary Education	0.249	0.222	0.045	0.250	0.214	0.060
Disabled	0.134	0.066	0.162	0.131	0.064	0.160
Single	0.211	0.265	-0.090	0.218	0.292	-0.120
Divorced	0.086	0.105	-0.046	0.088	0.106	-0.044
Widowed	0.068	0.026	0.143	0.070	0.025	0.150
East Germany	0.239	0.155	0.150	0.236	0.154	0.147
Retired	0.228	0.055	0.364	0.240	0.056	0.380
Maternity Leave	0.019	0.027	-0.037	0.017	0.023	-0.029
Unemployed	0.043	0.065	-0.069	0.042	0.064	-0.069
Non Working	0.080	0.077	0.007	0.075	0.074	0.001
In Education	0.026	0.042	-0.063	0.027	0.045	-0.071
Other Non Working	0.037	0.036	0.007	0.038	0.040	-0.009
Ln(Net HH Income)	7.872	7.879	-8.756	7.899	7.901	-2.476
Migration Background	0.117	0.291	-0.313	0.116	0.279	-0.296

Table 1: Descriptive statistics by treatment status

Notes: The table presents descriptive statistics for 2014 and 2015, by treatment status. The treatment group is composed by all respondents interviewed from 1 June until the end of the year, in 2015. The control group is composed by all respondents interviewed between 1 January and 30 May in 2015. The normalised difference is calculated as $\frac{\overline{X}_1 - \overline{X}_0}{\sqrt{S_0^2 + S_1^2}}$, where \overline{X}_1 is the mean of the control group, \overline{X}_0 is the mean of the treatment group, and S_1^2 and S_0^2 are the respective variances.

2015, and the full sample (2012–2015) is utilised to test the credibility of the identifying assumption. Hence, the main subsample is an unbalanced panel of 48,720 observations (23,532 for 2014 and 25,188 for 2015).

Table 1 reports descriptive statistics by treatment status, for both 2014 and 2015. The analysis includes a series of demographic controls (gender, age, education, marital status, migration background, disability), and labour market characteristics (net household income and working status). The table includes averages of the dependent variable and of all the control variables, along with the normalised differences between treatment and control group.¹³ As a rule of thumb, normalised differences exceeding a

¹³The normalized difference is the difference in averages by treatment status, scaled by the square root of the sum of the variances: $\frac{\overline{X}_1 - \overline{X}_0}{\sqrt{S_0^2 + S_1^2}}$.

quarter indicate a sample which is unbalanced on observable characteristics and might lead to sensitive results (Imbens and Wooldridge, 2009). The sample is unbalanced on age, migration background, and retirement status, with the treatment group including a younger subsample, a lower share of retired individuals and a higher share of individuals with a migration background. Nevertheless, I control for all these demographic characteristics.

3.2 Concerns: Preference over, or Salience of, Immigration?

Most of the studies in the economic literature of public opinion on immigration analyse measures of preferences, inquiring whether individuals would prefer more or less immigration. However, a recent paper by Hatton (2017a) stresses the need to account for the salience of immigration, defined as the importance which is attached by individuals to immigration, relative to other policy issues.

In his paper, preferences are measured utilising data from three survey questions of the European Social Survey (ESS).¹⁴ These are the following: "To what extent do you think Germany should allow people of the same race or ethnic group as most German people to come and live here?"; "How about people of a different race or ethnic group from most German people?"; "How about people from the poorer countries outside Europe?". The answers can be "Allow many to come and live here", "allow some", "allow a few", and "allow none". For all three questions, a measure of anti-immigration preferences is built with a dummy variable equal to one if the response is "allow a few" or "allow none", and zero otherwise. A comprehensive measure is composed of the average of the three questions.

A measure of salience is produced with data from the Standard Eurobarometer surveys.¹⁵ Each round, respondents are asked the following question: "What do you think are the two most important issues facing Germany at the moment?" Respondents can answer a maximum of two issues out of a list of fourteen.¹⁶ A measure of salience is created with a dummy variable equal to one if "Immigration" is mentioned by the respondent, and zero otherwise.

¹⁴The European Social Survey is a cross-national survey measuring attitudes, beliefs and behavior patterns of individuals in Europe. It is cross-sectional and conducted biannually since 2002, with 8 rounds of data (2002–2016), containing around 2,000 observations per country each round.

¹⁵The Standard Eurobarometer is a cross-national public opinion study operated on behalf of the European Commission since 1973. It is a cross-sectional survey conducted twice a year, in a spring and autumn round.

¹⁶In the latest rounds, these are "Crime", "Economic situation", "Rising prices/inflation", "Taxation", "Unemployment", "Terrorism", "Housing", "Government debt", "Immigration", "Health and social security", "The educational system", "Pensions", "The environment, climate and energy issues" or "Other".

The dependent variable I analyse in this paper is quite different from these two measures of preference and salience, but it might be correlated to them. Hence, understanding where a measure of concerns fits in this categorization becomes a relevant question, so as to grasp which component of public opinion the variable is picking up.

Prima facie, one would be inclined to consider concerns merely a measure of salience. In this perspective, being very concerned about immigration to Germany would be interpreted as attaching utmost importance to this specific policy issue. And this interpretation would hold irrespective of the preferences an individual has over immigration – be they more restrictive or permissive. However, the SOEP question measuring concerns is different from the salience question of the Eurobarometer in two ways. First, respondents of the SOEP questionnaire do not have to choose two important issues facing their country out of a list of fourteen, as in the Eurobarometer. Thus, for instance, when unemployment becomes a pressing issue, the salience of immigration goes down, whereas concerns about it might stay stable or even increase. Second, the framing of the question is not the same, in that being "very concerned" is somewhat different from "important issue facing our country".

An alternative view, which has been pushed forward in the SOEP literature, is implicitly considering concerns a measure of pure anti-immigration preferences, or attitudes.¹⁷ In this case, being very concerned about immigration would be interpreted as being against further immigration. However, it is also possible that an individual with a permissive attitude toward immigration is very concerned about it, or about the way the phenomenon is being handled by the government.

To shed some light on these alternative interpretations, I analyse the association of my measure of concerns with measures of preferences over, and salience of, immigration to Germany.¹⁸ Ideally, one would want to analyse measures of preferences, salience and concerns at the individual level in a panel setting. Unfortunately, to the best of my knowledge, such a dataset does not exist. Thus, I line up yearly measures from the three datasets at my disposal. Figure 4 presents such an exercise. For Eurobarometer data, the autumn and spring rounds are averaged to yield a yearly measure. SOEP data are provided yearly. Finally, ESS surveys are conducted every two years, and the average of the three questions is presented. Looking at the figure, it seems that concerns (in orange) closely follow preferences (in grey), until salience (in blue) increases. The correlation between salience and concerns is 0.2067. Instead, averaging concerns every two years to match the ESS data, the correlation between preferences and concerns is 0.6501.

¹⁷Mayda (2006) uses the terms "preference" and "attitude" interchangeably.

¹⁸I thank Timothy Hatton for this suggestion.



Figure 4: Salience, concern and preference over immigration in 2002-2017

Source: Eurobarometer (2017), SOEP (2015), European Social Survey (2017).

Notes: The blue line represents the share of respondents who report "Immigration" as one of the two most important issues facing Germany at the moment, in the Eurobarometer survey. The orange line represents the yearly average of the share of respondents who are very concerned about immigration to Germany in the SOEP dataset. Finally, the grey line represents the average value of three variables on immigration preference in Germany from the European Social Survey. The variables are constructed based on the following questions: 'To what extent do you think Germany should allow people of the same race or ethnic group as most German people to come and live here?"; "How about people of a different race or ethnic group from most German people?"; "How about people from the poorer countries outside Europe?" The variables measure the share of respondents who answer "allow a few" or "allow none". Moreover, utilizing SOEP data, a recent paper provides evidence about the intergenerational correlation of the measure of concerns I utilize in this work (Avdeenko and Siedler, 2017). Young adults, whose parents were ever very concerned about immigration during their childhood, have a higher likelihood of also ever being very concerned about it. This association lends some support to the view that concerns include a component of values that gets transmitted either from parents to offspring, or by the environment.

Taken together, this evidence suggests that my measure of concerns from the SOEP, although associated to both, is more correlated to a measure of preferences than to a measure of salience, as represented by ESS and Eurobarometer data respectively.

4 Empirical Strategy

As argued above, starting from June 2015, the peak of the refugee crisis took place, as the number of arrivals increased sharply and surpassed the number of asylum applications. In 2015, after 1 June, all individuals were treated by this event (the treatment group), and before that date all individuals were not treated (the control group). In 2014, both treatment and control groups were not treated. Exploiting the exogenous variation in survey interview timing of the SOEP data, I can estimate the causal effect of the treatment in a quasi-experimental setting, by comparing the outcomes of the treatment and control groups, in the time before (2014) and after (2015) the treatment. In other words, I employ a difference-in-differences strategy, analysing the change, between 2015 and 2014, in concerns about immigration of the treatment group, composed by individuals interviewed from June 2015 until the end of the year, and comparing it with the same difference for the control group, composed by all individuals interviewed between 1 January and 30 May 2015.

In the main analysis, the estimation equation is the following:

$$Y_{ismt} = \alpha + \beta_1 RefugeeCrisis_{ism} + \beta_2 Year2015_t + \beta_3 (RefugeeCrisis_{ism} * Year2015_t) + \gamma' X_{ismt} + \delta_s + \lambda_m + \mu_i + \varepsilon_{ismt}$$
(1)

where subscripts refer to individual i, residing in German state s, interviewed in month m of year t, the time indicator. Y is the dependent variable, a dummy variable equal to one if the individual is very concerned about immigration to Germany, and zero otherwise. RefugeeCrisis is a dummy equal to one if the individual was interviewed between 1 June and 10 November in 2015, and zero otherwise.¹⁹ This variable defines

¹⁹The last interview contained in the dataset was conducted on 10 November.

the treatment status based on the survey interview date in 2015, independent of when the individual was interviewed in other years. Year2015 is a dummy variable equal to one if the interview was conducted in 2015, and zero otherwise. X is a vector containing a variety of individual controls, such as gender, age, education, marital status, disability, working status, net household income and migration background. δ_s is a set of German state fixed effects, λ_m of month of interview fixed effects, and μ_i of individual fixed effects. Finally, ε_{ismt} is an error term.

The empirical strategy hinges on the crucial identifying assumption that, in the absence of the peak of the refugee crisis and the permissive government's asylum policy, the outcomes of treatment and control group would have followed parallel trends. Below, the credibility of this assumption is checked by showing that treatment and control group had a common trend in the years before the refugee crisis. Additionally, I run a series of regressions with the full sample of 2012–2015, containing placebo treatment interactions and finding no significant effect in the years prior to 2015. Hence, under the common trend assumption, the coefficient of interest, β_3 , yields the causal effect of the refugee crisis. The equation is estimated by pooled OLS and, alternatively, by individual fixed effects, to control for time-invariant unobserved heterogeneity. In both cases, standard errors are clustered at the individual level.

5 Results

In this section, I present the results of my empirical analysis. First, in Section 5.1, I find that the refugee crisis increased concerns about immigration to Germany. Moreover, I provide evidence supporting the credibility of the common trend assumption. In Section 5.2, I provide robustness checks, showing my main results are robust to alternative specifications. Finally, in Section 5.3, I test the heterogeneity of the treatment effect, finding that concerns increased more in East Germany.

5.1 Main Results

Table 2 presents the results of the main regressions based on equation (1). In the first two columns, I estimate the effect of the refugee crisis on concerns about immigration without controls, while the last two columns are estimated including all covariates, German state fixed effects and month of interview fixed effects. Columns (1) and (3) are estimated by pooled OLS, and columns (2) and (4) by individual fixed effects. The dependent variable is always a dummy equal to one if the individual is very concerned about immigration to

	(1)	(2)	(3)	(4)
	Very cond	cerned about	immigration to	Germany
	OLS	FE	OLS	FE
Year 2015	0.0453***	0.0462***	0.0471***	0.0467***
	(0.00349)	(0.00350)	(0.00389)	(0.00409)
Refugee Crisis	-0.00614		-0.00620	
	(0.00615)		(0.00777)	0 0 - 1 - + + + +
Refugee Crisis x Year 2015	0.0547^{***}	(0.0552^{***})	0.0575^{***}	0.0545^{***}
Ago	(0.00074)	(0.00085)	0.00315***	(0.00899)
Age			(0.00313)	
Age Squared			-0.00302***	
rige squared			(0.00117)	
Male			-0.00107	
			(0.00515)	
Secondary Education			-0.0289***	-0.0773*
			(0.00785)	(0.0451)
Tertiary Education			-0.164^{***}	-0.0835
			(0.00874)	(0.0583)
Single			-0.0231***	0.00428
			(0.00822)	(0.0343)
Divorced			-0.00276	0.00983
TT7-1			(0.00931)	(0.0414)
Widowed			-0.0156	(0.0208)
Ratirad			0.00387	0.0615***
netneu			(0.0123)	(0.0013)
Maternity Leave			0.0415**	0.0257
			(0.0162)	(0.0190)
Unemployed			0.0326***	4.06e-05
			(0.0118)	(0.0177)
Non Working			-0.00282	-0.00478
			(0.00992)	(0.0170)
In Education			-0.0864***	-0.0222
			(0.0165)	(0.0220)
Other Non Working			-0.0378***	0.00820
			(0.0125)	(0.0168)
Disabled			(0.0457^{****})	(0.0277)
Not HH Incomo			0.0534***	0.00242
ivet init income			(0.00513)	(0.0126)
Migration Background			-0.0510***	(010120)
gration Zaonground			(0.00719)	
Constant	0.255***	0.253***	0.644***	0.273**
	(0.00340)	(0.00157)	(0.0640)	(0.135)
Observations	48,720	48,720	44,042	44,042

Table 2: The refugee crisis and concerns about immigration

Notes: Refugee Crisis is equal to one if the individual was interviewed from 1 June until the end of the year in 2015, and zero otherwise. Year 2015 is equal to one if the individual was interviewed in 2015, and zero otherwise. Reference groups are year 2014, female, primary education, married, employed, not disabled, not with a migration background. Regressions in column (3) and (4) include German state and month of interview fixed effects. In regressions in column (2) and (4), Age and Age Squared are substituted by age dummies. Standard errors, clustered at the individual level, in parentheses. *** p<0.01; ** p<0.05; * p<0.1.

	(1)	(2)	(3)	(4)
	Very conce	rned about i	mmigration t	o Germany
Refugee Crisis	-0.0159^{*} (0.00953)		$\begin{array}{c} 0.00672 \\ (0.0102) \end{array}$	
Year2013	$\begin{array}{c} 0.0281^{***} \\ (0.00355) \end{array}$	$\begin{array}{c} 0.0287^{***} \\ (0.00353) \end{array}$	0.0296^{***} (0.00373)	$\begin{array}{c} 0.0288^{***} \\ (0.00374) \end{array}$
Year2014	$\begin{array}{c} 0.0693^{***} \\ (0.00364) \end{array}$	$\begin{array}{c} 0.0792^{***} \\ (0.00366) \end{array}$	0.0755^{***} (0.00383)	0.0778^{***} (0.00397)
Year2015	$\begin{array}{c} 0.115^{***} \\ (0.00385) \end{array}$	$\begin{array}{c} 0.125^{***} \\ (0.00389) \end{array}$	$\begin{array}{c} 0.123^{***} \\ (0.00410) \end{array}$	$\begin{array}{c} 0.125^{***} \\ (0.00443) \end{array}$
Refugee Crisis x Year 2013	$0.00855 \\ (0.0105)$	0.00874 (0.0105)	0.00688 (0.0111)	$0.00730 \\ (0.0111)$
Refugee Crisis x Year 2014	0.00972 (0.00996)	$\begin{array}{c} 0.00346 \\ (0.0103) \end{array}$	-0.0160 (0.0106)	$\begin{array}{c} 0.00472 \\ (0.0109) \end{array}$
Refugee Crisis x Year 2015	$\begin{array}{c} 0.0644^{***} \\ (0.0103) \end{array}$	$\begin{array}{c} 0.0593^{***} \\ (0.0106) \end{array}$	$\begin{array}{c} 0.0391^{***} \\ (0.0116) \end{array}$	0.0558^{***} (0.0120)
Estimation	OLS	\mathbf{FE}	OLS	\mathbf{FE}
Demographic controls	No	No	Yes	Yes
Labour market controls	No	No	Yes	Yes
State FE	No	No	Yes	Yes
Month of interview FE	No	No	Yes	Yes
Observations	79,614	79,614	72,573	72,573

Table 3: The refugee crisis and concerns about immigration, 2012-2015

Notes: Refugee Crisis is equal to one if the individual was interviewed from 1 June until the end of the year in 2015, and zero otherwise. Year 2013 is equal to one if the individual was interviewed in 2013, and zero otherwise. Year 2014 is equal to one if the individual was interviewed in 2014, and zero otherwise. Year 2015 is equal to one if the individual was interviewed in 2015, and zero otherwise. The reference year is 2012. Standard errors, clustered at the individual level, in parentheses. *** p<0.01; ** p<0.05; * p<0.1.

Germany, and zero otherwise. Across all specifications, the coefficients on the interaction term between "Refugee Crisis" and "Year 2015" are positive and statistically significant. The point estimates are stable to the inclusion of controls and individual fixed effects, suggesting results are not sensitive to observables and time-invariant unobservables. For the treatment group, interviewed during the peak of the refugee crisis, I find that the share of individuals who are very concerned about immigration increased by 5–6 percentage points. Relative to the pre-refugee crisis baseline level of concerns, this is an increase of around 22%. Thus, the refugee crisis, and the related government's asylum policy, substantially increased concerns about immigration to Germany. Moreover, the estimated coefficients of the variable "Year 2015" are positive and statistically significant across all specifications, indicating an increase in concerns from 2014 to 2015 for the control group as well.

The pooled OLS estimation in column (3) includes as controls all demographic and labour market characteristics. The effect of age is positive at a decreasing rate, indicating an inverted U-shaped relationship with concerns. Consistent with the literature on anti-immigration preferences (Hatton, 2017a), the level of education is negatively correlated with concerns. Individuals with a tertiary education are 16 percentage points less likely to be very concerned about immigration. Being single, being in the process of getting an education, and having a direct migration background, are all negatively correlated with concerns. The logarithm of net household income is also included in the regression, yielding a negative coefficient. Finally, individuals who are unemployed or on maternity leave, and those who are disabled, are more likely to be very concerned about immigration.

In Table 3, I present the results of the analysis covering the period 2012–2015, including two additional years in the regressions to control for underlining time trends. I estimate equation (1), also adding year dummies and placebo interaction terms between "Refugee Crisis" and these dummies. No dummy and interaction term is inserted for 2012, which constitutes the reference year. Again, I run regressions with and without controls, both by pooled OLS and individual fixed effects. In the last two columns, coefficients for controls are not displayed for the sake of brevity.²⁰ Across all specifications, the interaction term between "Refugee Crisis" and "Year 2015" is positive and statistically significant. Point estimates are unchanged at 4–6 percentage points. As expected, the estimated coefficients of all the interaction terms between the dummy defining the treatment group, "Refugee Crisis", and all the years prior to 2015, are close to zero and statistically insignificant. These results lend strong support to the credibility of the

²⁰The sign of the coefficients is the same as in Table 2, with similar magnitudes.

common trend assumption.

5.2 Robustness

If what I identify with the interaction term between "Refugee Crisis" and "Year 2015" is the effect of the refugee crisis, I do not expect this treatment to affect other concerns but those related to immigration. In table 4, I estimate equation (1) with the extended sample 2012–2015, substituting my main dependent variable with dummies measuring other possible concerns of the German people. These alternative concerns are about "hostility towards foreigners or minorities in Germany", "economic development", "your own economic situation", "your health", "maintaining peace", and "crime in Germany". In column (1), I find that the refugee crisis increased concerns about hostility towards foreigners and minorities in Germany. Only the estimated coefficient of the interaction with "Year 2015" is positive and statistically significant (at the 10% confidence level), supporting the credibility of the common trend assumption for this outcome as well. This result is interesting in and of itself, and not unexpected. Indeed, during 2015 there was a surge in the number of violent acts against refugees. It is possible that concerns about hostility towards foreigners increased because individuals believed there were rising xenophobic tendencies in the country. An alternative interpretation is that Germans became more emphatic towards foreigners and minorities, thus becoming more concerned about hostility in their regards (Schüller, 2016). In column (5), analysing the effect of the refugee crisis on the share of individuals very concerned about maintaining peace. the coefficient of the interaction term for 2015 is negative and statistically significant. However, I also estimate positive and statistically significant coefficients for the interactions of 2013 and 2014, suggesting that, before the refugee crisis, treatment and control groups followed different trends for this outcome. Finally, for the other concerns, all the estimated coefficients of the interactions are close to zero and statistically insignificant. As expected, I do not find any effect of the refugee crisis on these variables.

In a second robustness check, I test whether my results are robust to the inclusion of survey weights. In Table 5, I run my baseline regressions with fixed effects both for the restricted sample of 2014–2015 and the extended one of 2012–2015, weighting the data with a variable equal to the mean of the weighing factors of all the years included in the analysis. The estimated coefficients of my variable of interest remain positive and statistically significant. In column (1), for the sample 2014–2015, the coefficient is equal to 3.7 percentage points, slightly smaller than the one previously estimated. I conclude that my results are robust to the inclusion of survey weights.

	(1)	(2)	(3)	(4)	(5)	(9)
			Very concerned a	lbout		
	Hostility to foreigners	Economic Development	Own economic situation	Own Health	Mantaining Peace	Crime in Germany
Year 2013	-0.0125^{***} (0.00402)	-0.0381^{***} (0.00408)	-0.0277^{***} (0.00339)	-0.0163^{***} (0.00351)	-0.0218^{***} (0.00442)	0.00307 (0.00430)
Year 2014	0.0149^{***} (0.00421)	-0.0983^{***} (0.00407)	-0.0402^{***} (0.00345)	-0.00275 (0.00363)	0.0509^{***} (0.00475)	0.0333^{***} (0.00451)
Year 2015	0.127^{***} (0.00488)	-0.0817^{***} (0.00433)	-0.0469^{***} (0.00364)	-0.00871^{**} (0.00394)	0.179^{***} (0.00520)	0.0883^{***} (0.00493)
Refugee Crisis x Year 2013	0.00293 (0.0121)	-0.00492 (0.0123)	0.00609 (0.0105)	0.00971 (0.0101)	0.0290^{**} (0.0130)	0.0128 (0.0130)
Refugee Crisis x Year 2014	0.00788 (0.0118)	-0.0133 (0.0115)	-0.00379 (0.0101)	0.00711 (0.00981)	0.0493^{***} (0.0131)	0.00183 (0.0122)
Refugee Crisis x Year 2015	0.0245^{*} (0.0131)	-0.00590 (0.0122)	-0.00431 (0.0107)	0.00903 (0.0105)	-0.0734^{***} (0.0142)	-0.0186 (0.0130)
Estimation	FE	FE	FE	FE	FE	FЕ
Demographic controls	Yes	Yes	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$
Labour market controls	Yes	Yes	Yes	$\mathbf{Y}_{\mathbf{es}}$	Yes	Yes
State FE Month of interview FE	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes
Observations	72,419	72,440	72,493	72,498	72,484	72,492
Notes: Refugee Crisis is ϵ and zero otherwise. Year is equal to one if the indiv was interviewed in 2015, i level, in parentheses. ***	equal to one if th 2013 is equal to or idual was intervi and zero otherwi p<0.01; ** p<0.0	te individual was i one if the individu ewed in 2014, and se. The reference 5; * p<0.1.	interviewed from 1 ial was interviewed I zero otherwise. Y year is 2012. Stai	June until th l in 2013, and ear 2015 is eq ndard errors,	ie end of the y zero otherwise ual to one if th clustered at th	rear in 2015, 2. Year 2014 a individual a individual

Table 4: Robustness - The refugee crisis and other concerns

	(1)	(2)	(3)	(4)	(5)
	Very	concerned a	bout immigra	ation to Germ	nany
Year 2013		$\begin{array}{c} 0.0211^{***} \\ (0.00553) \end{array}$			
Year 2014		$\begin{array}{c} 0.0679^{***} \\ (0.00582) \end{array}$			
Year 2015	$\begin{array}{c} 0.0547^{***} \\ (0.00591) \end{array}$	$\begin{array}{c} 0.125^{***} \\ (0.00655) \end{array}$	$\begin{array}{c} 0.0525^{***} \\ (0.00350) \end{array}$	0.0469^{***} (0.00418)	$\begin{array}{c} 0.0469^{***} \\ (0.00958) \end{array}$
Refugee Crisis				-0.00496 (0.00877)	-0.00496 (0.0126)
Refugee Crisis x Year 2013		$\begin{array}{c} 0.0221 \\ (0.0154) \end{array}$			
Refugee Crisis x Year 2014		$\begin{array}{c} 0.0134 \ (0.0147) \end{array}$			
Refugee Crisis x Year 2015	$\begin{array}{c} 0.0368^{***} \\ (0.0136) \end{array}$	$\begin{array}{c} 0.0529^{***} \\ (0.0170) \end{array}$		$\begin{array}{c} 0.0579^{***} \\ (0.00925) \end{array}$	$\begin{array}{c} 0.0579^{***} \\ (0.00726) \end{array}$
Post Speech x Year 2015			$\begin{array}{c} 0.160^{***} \\ (0.0195) \end{array}$		
Estimation	\mathbf{FE}	\mathbf{FE}	\mathbf{FE}	OLS	OLS
Weights	YES	YES	NO	NO	NO
Clustering	Individual	Individual	Individual	Household	State
Demographic controls	YES	YES	YES	YES	YES
Labour market controls	YES	YES	YES	YES	YES
State FE	YES	YES	YES	YES	YES
Month of interview FE	YES	YES	YES	YES	YES
Observations	43,788	72,085	44,042	44,042	44,042

Table 5: Robustness - Weights, alternative treatment and clustering

Notes: Refugee Crisis is equal to one if the individual was interviewed from 1 June until the end of the year in 2015, and zero otherwise. Post Speech is equal to one if the individual was interviewed from 1 September until the end of the year in 2015, and zero otherwise. Year 2013 is equal to one if the individual was interviewed in 2013, and zero otherwise. Year 2014 is equal to one if the individual was interviewed in 2014, and zero otherwise. Year 2015 is equal to one if the individual was interviewed in 2015, and zero otherwise. Year 2015 is equal to one if the individual was interviewed in 2015, and zero otherwise. In all regressions, the reference year is 2014, and for column (2) it is 2012. Standard errors, clustered at the individual level, in parentheses. *** p<0.01; ** p<0.05; * p<0.1.

In column (3) of the same table, I estimate the baseline regression by changing the definition of my treatment group. In this specification, the treatment group is determined by the variable "Post Speech", a dummy equal to one for all individuals interviewed after 31 August, the day of Merkel's speech ("We can do it"), and zero otherwise. This is an alternative way to define my treatment, by catching up the effect of the very peak of the refugee crisis, starting from September 2015. I estimate an increase in the share of very concerned individuals of 16 percentage points, representing an increment of 53% compared to the pre-crisis baseline level of concerns.

Finally, in columns (4) and (5), I estimate the baseline regression by pooled OLS with clustering at the household and state level, instead of at the individual one. The estimated coefficients of interest are still significant at the 1% confidence level. I conclude my results are robust to these alternative specifications as well.

5.3 Effect Heterogeneity

The effect of the refugee crisis on concerns about immigration might have not been homogenous across the whole population. Individuals with different demographic characteristics may have reacted more or less strongly to the refugee crisis, and to the government's asylum policy. I test the heterogeneity of the treatment effect by adding triple interaction terms between "Refugee Crisis", "Year 2015" and some demographic characteristics of interest. Regressions are estimated with the 2014–2015 sample, by pooled OLS with clustering at the individual level, and not by fixed effects, because these demographic variables are mostly time-invariant. All regressions include the full set of controls but, in Table 6, I only display the estimated coefficients of interest.

First, I test effect heterogeneity with respect to the level of education. In column (1), the triple interaction term with "Tertiary Edu" represents the additional effect of the refugee crisis for the group of individuals who hold a tertiary education. These individuals on average are 16 percentage points less likely to be very concerned about immigration, and incurred an increase in concerns due to the refugee crisis 2 percentage points lower than that of the rest of individuals with a lower level of education. However, the coefficient of the triple interaction is only significant at the 10% level. This result is consistent with Schüller (2016), who finds that education has a moderating role on the effect of the 9/11 attack on concerns about immigration to Germany.

Second, I test whether the gender of individuals matters. The regression in column (2) shows that men incurred an increase in concerns 2 percentage points lower than that of women, although the coefficient is only significant at the 10% confidence level.

	(1)	(2)	(3)	(4)	(5)
	Very	concerned	about immigr	ation to Gerr	nany
Refugee Crisis x Year 2015	0.0622***	0.0669***	0.0495***	0.0620***	0.0535***
Refugee Crisis x Year 2015 x Tertiary Edu	(0.00907) -0.0216* (0.0197)	(18600.0)	(0.00000)	(02600.0)	(0.00964)
Refugee Crisis x Year 2015 x Male	(1710.0)	-0.0213*			
Refugee Crisis x Year 2015 x East Germany		(7110.0)	0.0594^{***}		
Refugee Crisis x Year 2015 x Mig Back			(2010.0)	-0.0157	
Refugee Crisis x Year 2015 x Age Over 45				(0.0134)	0.00715
Tertiary Edu	-0.161^{***}				(8110.0)
Male	(1.6800.0)	0.00234			
East Germany		(01:00040)	0.0411^{***}		
Mig Back			(U.UU0099)	-0.0471***	
Age Over 45				(60,000)	0.0119^{*} (0.00652)
Estimation	OLS	SIO	SIO	OLS	OLS
Demographic Controls	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	Yes
Labour Market Controls	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}
State FE	\mathbf{Yes}	\mathbf{Yes}	No	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$
Month of interview FE	Yes	Yes	Yes	\mathbf{Yes}	Yes
Observations	44,042	44,042	44,042	44,042	44,042

Heterogeneity	
Effect	
6:	
Table	

Third, in column (3), I investigate whether concerns have increased differentially in East and West Germany. The full set of German state fixed effects is substituted by a dummy equal to one if an individual lives in East Germany, and zero if he lives in the West. East Germans are on average 4 percentage points more likely to be very concerned about immigration and incurred an increase in concerns due to the refugee crisis of an additional 6 percentage points compared to West Germans. This means concerns about immigration of individuals living in East Germany increased more than double that those of individuals living in the West. The refugee crisis and the related government's asylum policy had a stronger effect in increasing people's concerns in East Germany. This result might be indicative of a persistent difference in the cultural and social histories of the two parts of the country.

Forth, I test whether individuals with a direct migration background were less affected by the refugee crisis. In column (4), I find there is no evidence of a heterogeneous treatment effect with respect to migration background, as the coefficient of the interaction term is negative, but statistically insignificant. Finally, I find no evidence of a heterogeneous treatment effect with respect to age. I test this by inserting a dummy equal to one for individuals over 45 years old, instead of the usual age controls. The triple interaction term is close to zero and statistically insignificant.

6 Concerns about Immigration and Support for the AfD

As argued above, the refugee crisis substantially increased German concerns about immigration. Clearly, this increase in concerns may have played a role in shifting political preferences, to the benefit of the recently established populist, right-wing party, *Alternative für Deutschland* (AfD). Indeed, the AfD was founded in February 2013 as a single-issue party, criticizing the Euro, and more generally the European Union. In the federal election of 2013, the party gained 4.7% of the vote, reaching a near-success in such a short time since its founding, but missing the threshold of 5% to enter the parliament. After the election, the AfD began shifting its focus from the Euro crisis to the pressing issue of immigration (Schmitt-Beck, 2017). In this respect, the AfD adopted populist communication strategies, promoting a nationalist and conservative stance. Once the refugee crisis broke out in the summer of 2015, the party further stressed its strong antiimmigration position and criticized the welcoming asylum policy enacted by Chancellor Merkel.

In the federal election of 2017, the AfD received 12.6% of the vote, becoming the third largest party in the German parliament. Moreover, the party gained a stronger

electoral support in East Germany, for instance, claiming 27% of the vote in the state of Saxony. Importantly, immigration was the single most important issue for the German population during the election (Dostal, 2017).²¹ Thus, concerns about immigration might have played a crucial role in shifting political support of voters towards the AfD.

Unfortunately, SOEP data regarding voting in 2017 are not available yet. Hence, I resort to analysing the relationship between concerns about immigration and party support for 2014 and 2015, the only two years in which data for the AfD support are available. Respondents are asked whether they support a political party.²² In case of affirmative answer, they are then asked the following question: "Which party do you lean toward?". I create a dummy variable equal to one if an individual supports the AfD, and zero if it supports another party. Similarly, I create dummy variables for each party present in the German electoral field.

In Panel A of Table 7, I regress these measures of support for parties on a dummy variable equal to one for individuals who are "very concerned about immigration to Germany", and zero otherwise. All regressions include the full set of demographic and labour market controls, German state fixed effects, month of interview fixed effects and are estimated by individual fixed effects, with clustering at the individual level. The estimated coefficient for the AfD is positive and statistically significant. Individuals who are very concerned about immigration are 0.88 percentage points more likely to support the AfD. This is a substantial effect, because, in my dataset, only 2.21% of respondents support the AfD in 2014 and 2015, conditional on declaring which party they support. The estimated coefficients for all the other parties are close to zero and statistically insignificant.

However, it is important to note these regressions estimate correlations, and not causal effects. Indeed, the usual endogeneity issues are likely to be at play. Omitted variables might account for both concerns about immigration and political support for the AfD. Moreover, causality may be running in the other direction. It is possible that an individual who supports the AfD becomes more concerned about immigration by being affected by the xenophobic rhetoric supplied by the party. This hypothesis is supported by a recent paper of Müller and Schwarz (2017), who find that anti-refugee hate speech on the Facebook page of the AfD predicts violent crimes against refugees in Germany.

Next, I attempt to estimate the casual effect of concerns about immigration on

²¹This is confirmed by the Standard Eurobarometer, autumn round of 2017. Immigration was cited as the most important issue facing the country by 40% of respondents, followed by the education system (22%), terrorism (20%), and the environment, climate and energy issues (20%).

 $^{^{22}}$ In my dataset, 41.9% of respondents in 2015 support a political party. Of these, 95.2% report which specific party they support.

CSU FDP 3 -0.00296 -0.00340 3) (0.00439) (0.00275)	CDD				
$\begin{array}{rrrr} 3 & -0.00296 & -0.00340 \\ 3) & (0.00439) & (0.00275) \end{array}$		Greens	Linke	NPD	Pirate Party
	0.00436 (0.00613)	0.000671 (0.00386)	-0.00230 (0.00360)	-0.00158 (0.00119)	0.000841 (0.00190)
Yes Yes	Yes	Yes	Yes	Yes	Yes
Yes Yes	Yes	Yes	$\mathbf{Y}_{\mathbf{es}}$	Yes	Yes
Yes Yes	Y_{es}	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	Yes
Yes Yes	Yes	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	Yes
Yes Yes	Yes	Yes	\mathbf{Yes}	Yes	Yes
8 19,128 19,128	19,128	19,128	19,128	19,128	19,128
dummy equal to one if the if the individual was inte ridual was interviewed in 20 scific party, and zero if the	e individual is erviewed from 215, and zero individual su	very concerr 1 June unt otherwise. D pports any a	ned about in il the end of Dependent va unother party	amigration t the year in riables for p v. Standard o	o Germany, and 2015, and zero arty support are errors, clustered
Yes Yes Ye Yes Ye Yes Ye Yes $19,12$ 19,128 $19,1dummy equal to one ifdummy equal to one ifif the individual wasridual was interviewed iscific party, and zero if0.01; ** p<0.05; * p<0$	s s s 28 .: the the in 26	s Yes s Yes s Yes s 19,128 28 19,128 the individual is interviewed from in 2015, and zero the individual su	s Yes Yes Yes s Yes Yes Yes s Yes Yes S 19,128 19,128 the individual is very concer- interviewed from 1 June unt in 2015, and zero otherwise. I the individual supports any a	s Yes Yes Yes Yes Yes Yes S s Yes Yes Yes Yes Yes s 19,128 19,128 19,128 the individual is very concerned about in interviewed from 1 June until the end of in 2015, and zero otherwise. Dependent va the individual supports any another part.	s Yes Yes Yes Yes Yes Yes Yes S s Yes Yes Yes Yes Yes Yes Yes Yes Yes Ye

AfD
or the
support fo
and
ration
immig
about
Concerns
Table

support for the AfD by exploiting the exogenous variation in concerns generated by the refugee crisis. Under the assumption that the refugee crisis, and the related government's asylum policy, affected AfD support only through concerns about immigration, I can employ an instrumental variable strategy. Since this condition may not be credible, I consider the following analysis as tentative. First, in Panel B of Table 7, I estimate the reduced form by regressing equation (1), substituting the usual dependent variable with support for the AfD. Surprisingly, I find that, during the peak of the refugee crisis, support for the AfD decreased, although the effect is not statistically significant. In Panel C, I estimate the effect of concerns on support for the AfD with 2SLS estimation. Again, the estimated coefficient is negative and statistically insignificant. Thus, it seems that the correlation between concerns about immigration and support for the AfD is not of a causal nature. If anything, these results suggest that support for the AfD decreased during the refugee crisis.

However, these findings are subject to a series of limitations. First, as already noted, the identifying assumption of the instrumental variable strategy might be violated. Second, the reduced form estimates are biased if the common trend assumption is not respected for the variable measuring AfD support. Unfortunately, there is no way to check the credibility of this assumption directly, because the AfD was founded in 2013 and I only dispose of data concerning support for the party in 2014 and 2015. Third, and more importantly, I do not analyse a measure of voting, but rather of political preference. Thus, it is possible that individuals do not fully disclose their real preferences, especially when it comes to declaring support for a right-wing populist party that expresses radical and xenophobic positions. Alternatively, it might be the case that the AfD, relative to other parties, holds a lower support among the electorate that the share of votes it ends up getting at the elections. In other words, it is possible that on election day the AfD succeeds, more than its competitors, in mobilizing parts of the electorate that do not steadily support the party. Indeed, my analysis focuses on the short term, in that I attempt to find an effect of concerns on political support in the second part of 2015. However, it is probable that concerns about immigration affected political outcomes in the following period, and precisely on the election day.

In sum, I find that concerns about immigration are positively correlated with support for the AfD. This correlation suggests that AfD supporters are characterised by some personality or cultural traits that lead them to be more concerned about immigration. Exploiting the exogenous variation in concerns generated by the refugee crisis, I provide tentative evidence suggesting that this correlation does not have a causal nature. If anything, I find that support for the AfD diminished during the peak of the refugee crisis. More research is needed to establish if the increase in concerns due to the refugee crisis caused an electoral gain for the AfD in the federal election of 2017.²³

7 Conclusion

The refugee crisis represented one of the most important events in contemporary German history. In this paper, exploiting exogenous variation in survey interview timing of the SOEP, I estimate the short-term causal effect of this considerable influx of asylum seekers and migrants on concerns about immigration. The estimated effect is substantial, representing an increase of around 22%, compared to the pre-refugee crisis baseline level of concerns. Interestingly, I find that this increase was twice as large for East Germans, compared to West Germans. In a second section, I show concerns about immigration are positively correlated with political support for the relatively new, right-wing populist party *Alternative für Deutschland*. However, using the variability in concerns generated by the refugee crisis, I find no evidence of a causal effect of concerns on political preferences in the short term.

My findings suggest that the refugee crisis, and the related government's asylum policy, affected public opinion on immigration in Germany, by substantially increasing the share of people who are very concerned about immigration. Refugee flows might not only bring about large distributional consequences in the labour market (Borjas and Monras, 2017), but also affect people's anxieties. Moreover, I argue concerns about immigration may be an independent driver of populist voting, specifically for the case of Germany, which received the most refugees in Europe and was not affected much by the economic crisis. Future research is needed to investigate whether the increase in concerns generated by the refugee crisis affected political outcomes in favour of the AfD. For the moment, my paper makes a first step by uncovering the correlation between concerns about immigration and support for this right-wing populist party.

 $^{^{23}}$ Data on voting in 2017 are collected by the SOEP during 2018 and will likely be released in December 2019.

References

Algan, Y., S. Guriev, E. Papaioannou, and E. Passari (2017). "The European Trust Crisis and the Rise of Populism." *Brookings Papers on Economic Activity*, Fall.

Avdeenko, A. and T. Siedler (2017). "Intergenerational Correlations of Extreme Right-Wing Party Preferences and Attitudes toward Immigration." *Scandinavian Journal of Economics*, 119(3), 768–800.

BAMF (2016). "Das Bundesamt in Zahlen 2015. Asyl, Migration und Integration." Federal Office for Migration and Refugees.

Barone, G., A. D'Ignazio, G. de Blasio, and P. Naticchioni (2016). "Mr. Rossi, Mr. Hu and Politics. The Role of Immigration in Shaping Natives' Voting Behavior." *Journal of Public Economics*, 136, 1–13.

Boomgaarden, H. G. and R. Vliegenthart (2009). "How News Content Influences Anti-Immigration Attitudes: Germany, 1993–2005." *European Journal of Political Research*, 48, 516–542.

Borjas, G. J. and J. Monras (2017). "The Labour Market Consequences of Refugee Supply Shocks." *Economic Policy*, 32(91), 361–413.

Caliendo M. and K. Wrohlich (2010). "Evaluating the German 'Mini-Job' reform using a natural experiment." *Applied Economics*, 42(19), 2475–2489.

Card, D., C. Dustmann and I. Preston (2012). "Immigration, Wages, and Compositional Amenities." *Journal of the European Economic Association*, 10, 78–119.

Ceobanu, A. M. and X. Escandell (2010). "Comparative Analyses of Public Attitudes Toward Immigrants and Immigration Using Multinational Survey Data: A Review of Theories and Research." *Annual Review of Sociology*, 36, 309–28.

Citrin, J., D. P. Green, C. Muste and C. Wong (1997). "Public Opinion toward Immigration Reform: The Role of Economic Motivations." *Journal of Politics*, 59(3), 858–81.

De Poli, S., N. Jakobsson and S. Schüller (2017). "The Drowning-Refugee Effect: Media Salience and Xenophobic Attitudes." *Applied Economics Letters*, 24(16), 1167–1172.

Doerrenberg, P. and S. Siegloch (2014). "Is Soccer Good for You? The Motivational Impact of Big Sporting Events on the Unemployed." *Economics Letters*, 123(1), 66–69.

Dostal, J. M. (2017). "The German Federal Election of 2017: How the Wedge Issue of Refugees and Migration Took the Shine off Chancellor Merkel and Transformed the Party System." *The Political Quarterly*, 88, 589–602.

Dustmann, C. and I. Preston (2007). "Racial and Economic Factors in Attitudes to Immigration." The B. E. Journal of Economic Analysis and Policy, 7, 1655–1655.

Dustmann, C., K. Vasiljeva, and A. P. Damm (2016). "Refugee Migration and Electoral Outcomes." CReAM Discussion Paper Series, 1619.

Dustmann, C., F. Fasani, T. Frattini, L. Minale, and U. Schnberg (2017a). "On the Economics and Politics of Refugee Migration." *Economic Policy*, 32(91), 497–550.

Dustmann, C., B. Eichengreen, S. Otten, A. Sapir, G. Tabellini, and G. Zoega (2017b). "Europe's Trust Deficit: Causes and Remedies." CEPR Press.

Facchini, G. and A. M. Mayda (2008). "From Individual Attitudes to Immigration Policy: Theory and Evidence." *Economic Policy*, 56, 651–713.

Facchini, G., A. M. Mayda, and R. Puglisi (2009). "Illegal Immigration and Media Exposure: Evidence on Individual Attitudes." CEPR Discussion Paper 7593.

Gidron, N. and B. Bonikowski (2013). "Varieties of Populism: Literature Review and Research Agenda." Harvard University, Weathgerhead Center for International Affairs. Working Paper 13.

Goebel, J., C. Krekel, T. Tiefenbach, and N. R. Ziebarth (2013). "Natural Disaster, Policy Action, and Mental Well-Being: The Case of Fukushima." IZA Discussion Paper 7691.

Guiso, L., H. Herrera, M. Morelli and T. Sonno (2017). "Populism: Demand and Supply." Working Paper.

Haaland, I. and C. Roth (2017). "Labor Market Concerns and Support for Immigration." Working Paper.

Hainmueller, J. and M. J. Hiscox (2007). "Educated Preferences: Explaining Individual Attitudes toward Immigration in Europe." *International Organization*, 61, 399–442.

Hainmueller, J. and D. Hopkins (2014). "Public Attitudes toward Immigration." Annual Review of Political Science, 17, 225–249.

Halla, M., A. F. Wagner, and J. Zweimüller (2017). "Immigration and Voting for the Far Right." *Journal of the European Economic Association*, 15(6), 1341–1385.

Hatton, T. J. (2016a). "Immigration, Public Opinion and the Recession in Europe." *Economic Policy*, 86, 205–246.

Hatton, T. J. (2016b). "Refugees, Asylum Seekers and Policy in OECD Countries." American Economic Review (Papers and Proceedings), 106, 441–445. Hatton, T. J. (2017a). "Public Opinion on Immigration in Europe: Preference Versus Salience?" CEPR Discussion Paper 12084.

Hatton, T. J. (2017b). "Refugees and Asylum Seekers, the Crisis in Europe and the Future of Policy." *Economic Policy*, 32(91), 447–496.

Imbens, G. W. and J. M. Wooldridge (2009). "Recent Developments in the Econometrics of Program Evaluation." *Journal of Economic Literature*, 47 (1), 5–86.

Inglehart R. F. and P. Norris (2016). "Trump, Brexit, and the Rise of Populism: Economic Have-Nots and Cultural Backlash." Harvard Kennedy School RWP16-026.

Jäckle, S. and P. D. König (2017). "The Dark Side of the German 'Welcome Culture': Investigating the Causes Behind Attacks on Refugees in 2015." *West European Politics*, 40(2), 223–251.

Mayda, A. M. (2006). "Who is Against Immigration? A Cross-Country Investigation of Attitudes Towards Immigrants." *Review of Economics and Statistics*, 88, 510–530.

Müller, K. and C. Schwarz (2017). "Fanning the Flames of Hate: Social Media and Hate Crime." Working Paper.

Otto, A. H. and M. F. Steinhardt (2014). "Immigration and Election Outcomes - Evidence from City Districts in Hamburg." *Regional Science and Urban Economics*, 45, 67–79.

O'Rourke, K. H. and R. Sinnott (2006). "The Determinants of Individual Attitudes Towards Immigration." *European Journal of Political Economy*, 22 (4), 838–861.

Rodrik, D. (2017). "Populism and the Economics of Globalization." CEPR Discussion Paper 12119.

Scheve, K. F. and M. J. Slaughter (2001). "Labor Market Competition and Individual Preferences over Immigration Policy." *Review of Economics and Statistics*, 83, 133–145.

Schüller, S. (2016). "The Effects of 9/11 on Attitudes toward Immigration and the Moderating Role of Education." *Kyklos*, 69, 604–632.

Schmitt-Beck, R. (2017). "The 'Alternative für Deutschland in the Electorate': Between Single-Issue and Right-Wing Populist Party." *German Politics*, 26(1), 124–148.

Singleton, A, A. Carassco-Heiermann and D. Kierans (2016). "Migration, Asylum and Refugees in Germany: Understanding the Data." IOM GMDAC Data Briefing Series, 1-6. Socio-Economic Panel (SOEP). Data for years 1984-2015, version 32, SOEP, 2016, doi: $10.5684/{\rm soep.v32}$

Wagner, G. G., J. R. Frick, and J. Schupp (2007). "The German Socio-Economic Panel Study (SOEP) – Scope, Evolution and Enhancements." *Schmollers Jahrbuch*, 127 (1), 139–169.